IZA DP No. 1430

Riches to Rags Every Month? The Fall in Consumption Expenditures Between Paydays

David Huffman Matias Barenstein

December 2004

Forschungsinstitut zur Zukunft der Arbeit Institute for the Study of Labor

Riches to Rags Every Month? The Fall in Consumption Expenditures Between Paydays

David Huffman

IZA Bonn

Matias Barenstein

U.S. Federal Trade Commission

Discussion Paper No. 1430 December 2004

IZA

P.O. Box 7240 53072 Bonn Germany

Phone: +49-228-3894-0 Fax: +49-228-3894-180 Email: iza@iza.org

Any opinions expressed here are those of the author(s) and not those of the institute. Research disseminated by IZA may include views on policy, but the institute itself takes no institutional policy positions.

The Institute for the Study of Labor (IZA) in Bonn is a local and virtual international research center and a place of communication between science, politics and business. IZA is an independent nonprofit company supported by Deutsche Post World Net. The center is associated with the University of Bonn and offers a stimulating research environment through its research networks, research support, and visitors and doctoral programs. IZA engages in (i) original and internationally competitive research in all fields of labor economics, (ii) development of policy concepts, and (iii) dissemination of research results and concepts to the interested public.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

IZA Discussion Paper No. 1430 December 2004

ABSTRACT

Riches to Rags Every Month? The Fall in Consumption Expenditures Between Paydays^{*}

This paper finds declining consumption expenditure between paydays, for a typical household in the working population of the UK. The magnitude is inconsistent with exponential time preference, but compatible with quasi-hyperbolic discounting. However, the hyperbolic model predicts that credit constraints drive the decline, and we find only mixed evidence in this regard. We also observe a method-of-payment result that suggests a role for mental accounting: households choose declining cash spending but flat credit-card spending over the pay period. We propose an alternative explanation for the results, based on cognitive costs of budgeting and perceptual biases, rather than self-control problems.

JEL Classification: B49, D11, D12, J33

Keywords: consumption, hyperbolic-discounting, payday, mental accounting, reference-dependent preferences, credit cards

Corresponding author:

David Huffmann IZA Bonn P.O. Box 7240 53072 Bonn Germany Email: huffmann@iza.org

^{*} Any opinions expressed here are those of the authors and not those of the institute or the U.S. Federal Trade Commission or of any individual Commission. We would like to thank George Akerlof, Matthew Rabin, Lorenz Goette, Kenneth Chay, Dan Ariely, Stefano Della Vigna, George Loewenstein, Michael Janson, Terrance Odean, Jim Ohls, Ernesto Dal Bó, Pipat Luengnaruemitchai, David Romer, James Wilcox, David Laibson, Barbara Mellers, Jonathan Zinman, David Laibson, Christian Geckeler, Marty Olney, David Jaeger, and participants of the UC Berkeley psychology and economics seminar for helpful comments and suggestions. We are also grateful to the staff from the UK Data Archive at the University of Essex for providing us with access to the FES survey (Crown Copyright) and for answering our questions; in particular, Nadeem Ahmad, Karen Dennison, Myriam Garcia Bernabe, Jack Kneeshaw, and Palvi Shah.

"Remember: If you don't see it, you won't spend it! ... If your company offers a 401(k) retirement plan, make sure you sign up for the maximum possible contribution. It will be taken out of your *paycheck* automatically... The whole point is to get the money out of your checking account before you see it and spend it."

- T. Savage, *How small cuts become huge savings*, MSN Money website (undated)

1. Introduction

This paper finds a pattern of declining consumption expenditure between paydays, for households in the general working population.¹ The magnitude of the decline is not trivial, amounting to roughly 12% of weekly household expenditure. Given that recipients generally know the amount and timing of paychecks well in advance, this sensitivity of consumption to receipt of the paycheck contradicts classic models of consumption smoothing.²

Assuming negative time-preference, and binding credit constraints, the standard model comes closer to capturing the qualitative features of the pattern we observe: impatience generates declining consumption over time, and binding credit constraints lead to a discrete jump in consumption after each payday. However, a calibration exercise produces the second main finding of the paper: the exponential daily discount rate assumed in the standard model cannot explain the *magnitude* of the decline we observe. The daily discount rate needed to explain the decline implies an implausibly high annual discount rate, or low annual discount factor, between 0.5 and 0.1.³

¹ We use the Family Expenditure Survey (Crown Copyright), a random national sample from the working population in the United Kingdom.

² The data reject several plausible explanations for non-smooth consumption: stockpiling, calendar-month effects, timing of non-discretionary payments, or intra-household strategic motives.

³ I.e. the consumer values consumption today about 70 percent more than consumption in one year's time, and 90 percent more than in two year's time. Estimates become even

We show in another calibration exercise that the decline is consistent with quasihyperbolic time preference, in which an individual has two discount rates, one for the short-run and one for longer time horizons. This finding is important because it is one of the few pieces of evidence from outside the laboratory on the daily discounting of households in the general working population.⁴ It is also important because the seemingly innocent assumption of a non-constant discount rate gives rise to a self-control problem, in the sense of dynamically inconsistent, "present-biased" preferences.

The hyperbolic model can explain the magnitude of the decline we observe, but is harder to reconcile with a fourth finding of the paper, a decline in expenditures even for households who own a credit card. As with any explanation based on time preference, the quasi-hyperbolic model requires binding credit constraints to explain a jump in expenditure after each payday. To the extent that households with credit cards have the ability to smooth short-term consumption, some other factor besides impatience is needed to explain the jump in spending after payday.

Another finding of the paper is an intriguing method-of-payment result. Households with credit cards choose declining cash expenditures, but "flat" credit card expenditures, over the course of the pay period. The decline in cash payments is comparable in magnitude to the decline for households without credit cards. Neither the standard model nor a model with hyperbolic discounting is sufficient to predict this

more implausible for longer time horizons, precisely because the discount rate is constant.

⁴ Shapiro (2004) finds a similar, although slightly, daily discount rate for food stamp recipients. DellaVigna and Paserman (2001) estimate the discount rates of health club members, DellaVigna and Malmendier (2002) do the same for job searchers, and Laibson, Repetto and Tobacman (2003) infer discount rates from lifecycle savings behavior.

pattern. We interpret this as evidence of a system of "mental accounting" at work, with different rules for spending out of the current bank account, as opposed to the flow of future income accessible with the credit card.⁵ One potential explanation for the decline we observe is thus hyperbolic discounting, combined with a mental accounting rule that to some extent limits credit card spending and causes households to act as though they are credit constrained.

After presenting the results, the paper considers alternative explanations that have not been discussed in the literature. One explanation corresponds to a commonly held intuition, that consumers have a tendency to feel that spending is "less costly" towards the beginning of the pay period, when there has been a salient increase in the bank balance. As a consequence of this feeling of being "flush," they overspend. Because this type of perceptual bias has some basis in evidence from psychology, e.g. evidence on reference-dependent preferences, and could generate a decline in spending between paydays, we sketch a simple model that captures this intuition. This model also offers a different interpretation of the method-of-payment result, suggesting that flat credit card expenditures reflect the feeling that it is always payday when one spends with a credit card. However, we conclude that more research is needed to disentangle the relative importance of impatience, and perceptual biases, in explaining monthly budgeting behavior.

In addition to its theoretical interest, the decline in consumption spending over the pay period may be of interest to policy makers. Economists criticize in-kind transfer programs, designed to regulate the monthly spending of low-income individuals, from the

⁵ See Thaler (1999) for an overview of the (mostly laboratory) evidence on mental accounting.

perspective of fully rational choice. Given that the decline in spending between paydays is difficult to explain without allowing for self-control problems or perceptual biases, these programs may be more-easily defended.⁶ Evidence of sub-optimal budgeting during the pay period has larger implications as well, adding support to the view that small mistakes in budgeting can add-up over the lifetime, leading to sub-optimally low saving for retirement. In this case policies designed to encourage retirement saving may be appropriate.⁷ Of course we must proceed with caution, because at this point it is hard to say exactly how large these welfare losses are. Furthermore, it is important that policies allow for the fact that some consumers may be less subject to self-control problems and biases than others.⁸

The paper is organized as follows. Section 2 discusses related literature, Section 3 describes the data and empirical design, and Section 4 presents the main results. Section 5 discusses the role of time preference and credit constraints, Section 6 presents our method-of-payment results, Section 7 discusses alternative explanations, and Section 8 concludes.

2. Related Literature

⁶ In unreported regressions, we find a large payday effect for spending on alcohol, raising welfare concerns relating to health costs and productivity loss due to binge drinking.
⁷ E.g. research on personal saving (Choi et al., 2001) has shown that employees save substantially more over time, if they participate in employer-sponsored defined contribution pension plans (such as 401(k)'s), which automatically debit part of their paychecks.

⁸ See Camerer et al (2003) for a discussion of "asymmetric paternalism," a policy approach designed to help less-than-rational decision makers, without unduly harming rational decision makers.

Several recent papers find an impact of payday on consumption spending.⁹ Stephens (2003) uses daily expenditure data from the CES, and finds that social security recipients spend significantly more right after the third of the month, when they receive their social security checks. Stephens concludes that this behavior violates the consumption-smoothing prediction of the Lifetime Permanent Income Hypothesis. Stephens remains largely silent regarding an explanation for what he finds, however, except to suggest that credit constraints could explain the phenomenon. Stephens (2004) is also conducting research on the decline in spending between paydays using the Family Expenditure Survey (FES). In this paper Stephens is again focused mainly on testing the LPIH, and proposes credit constraints as the main candidate explanation for the decline.

Shapiro (2004) uses daily caloric intake data on food stamp recipients in the US, and finds that recipients choose a pattern of declining consumption between monthly food stamp payments. Shapiro performs calibration exercises similar to the ones in this paper, showing that the magnitude of the decline cannot be explained with exponential time preference. Instead, Shapiro argues that hyperbolic discounting is the most likely explanation for the decline. Shapiro concludes that, while the decline may reflect suboptimal budgeting, it is too small to imply significant welfare losses.

This paper differs from Stephens (2003, 2004) and Shapiro (2004) in a number of ways. Unlike Stephens, we leverage the decline in spending following payday as a useful tool for testing between alternative models of consumer choice, including non-standard models from behavioral economics. We calibrate the standard and hyperbolic models in a

⁹ The initial version of this paper was completed before any of these related papers was publicly available. However, we have benefited from observing the literature evolve, and have endeavored to keep our research up-to-date and complementary.

similar way to Shapiro (2004), ruling-out credit constraints and exponential time preference as an explanation, but our results are more general than Shapiro (2004), as they apply to the working population rather than just food stamp recipients.¹⁰ Another advantage compared to Shapiro is the fact that our data includes individuals who are not credit constrained, allowing us to test whether the decline only happens for individuals facing binding credit constraints, a prediction of the hyperbolic model. We are also able to explore how the decline varies with method of payment, and study the role of mental accounting, which Shapiro and Stephens do not do. Finally, we formalize an alternative intuition regarding the source of the decline in spending, based on perceptual biases rather than impatience.

3. Description of the Data and Empirical Design

3.1 The Family Expenditure Survey

The Family Expenditure Survey (FES) is administered every year in the United Kingdom, to a sample of between six and seven thousand households. The survey does not follow the same households across years. The survey includes an interview with at least one household member, asking about demographic and household information. Crucially for this paper, individuals report the amount and date of their last paycheck, and the frequency of pay.

Following the day of the interview, each participating household member starts a diary, and records everything they buy during the next fourteen days. Expenditures are

¹⁰ A limitation of our results compared to Shapiro is that we use weekly data, and therefore must make an additional assumption to recover the daily discount rate. However, we find evidence that this assumption is in fact a reasonable one.

aggregated to diary "weeks" in the data, for reasons of confidentiality, resulting in two seven-day aggregates of expenditure for each individual. Importantly, the timing of the FES interview, and the diary recordings, is random during the sample year. Figure 1 illustrates the resulting data structure: diary weeks need not correspond to the calendar week, but rather start on different days of the week, at different distances from payday, and overlap to varying degrees.

The timing of the FES interview is important for a correct interpretation of the timing of payday. The interview occurs immediately before the diary period starts, so the question about the date of the most recent paycheck refers to a paycheck received before the diary period. Because the survey also includes the frequency of pay, e.g. calendar month, it is possible to impute the timing of the next payday. However, there is some measurement error involved in imputing the next payday, which is discussed in detail below.

3.2 Empirical Design

Estimating the weekly expenditure profile

The motivating question for the paper suggests a straightforward empirical design, in which expenditure is modeled as a function of distance from payday in days, over the course of a monthly pay period.¹¹ Because the FES diary data are aggregated over seven-day periods, we start by investigating how these week-long aggregates of

¹¹ The analysis focuses on a monthly pay period for reasons discussed below.

household expenditure change, as the start-date of the week gets farther away from the payday of the main earner.¹²

The basic regression we estimate is of the form:

$$C_{it} = \alpha + \sum_{t=1}^{3} \beta_t distanc \quad e_t + \gamma \cdot T_t + \eta \cdot Z_i + \varepsilon$$
(1)

The dependent variable, C_{it} , is the log of consumption expenditure by household *i*, during the diary week beginning at time *t*. The distance measure consists of four distance categories, represented by four dummy variables (the third category is omitted in the equation).¹³ The first category includes diary weeks starting on payday and weeks starting 1 to 7 days after payday. The second category includes weeks starting 8 to 14 days after payday, and the third category includes weeks starting 15 to 22 days after payday. We exclude from our analysis weeks beginning 23 days after payday during 30day pay periods (one day more or less for other lengths of pay period), but starting more than 4 days before the next payday. This is because measurement error makes it unclear whether these weeks include the next payday or not. Measurement error arises because employers might use one of two rules for the timing of a monthly payday: (1) pay on the same date of each month, or (2) always pay on the first or last day of the month.¹⁴ The fourth and final distance category includes weeks that are certain to include the next

¹² We will often refer to dairy weeks as "weekly expenditures," meaning only that these aggregates are one week long, not that they correspond to calendar weeks.

¹³ These categories correspond roughly to the four weeks of the pay period, and summarize the relationship observed in our graphical analysis. We verify that our basic results are also robust in a completely non-parametric specification, with 31 individual distance dummies.

 $^{^{14}}$ E.g. it is unclear whether someone paid on the 30th of April, the last day of the month, will next be paid on the 30th of May, or the 31st. Including these weeks would only obscure the relationship of interest, so we omit them. Stephens (2004), by contrast, uses all diary weeks.

(imputed) payday, i.e. weeks that begin 4 to 1 days before the next payday. We keep this category separate from weeks beginning on the next payday, in order to avoid confounding the payday effect with possible end-of-pay-period effects.¹⁵

The vector T_t includes dummies for day of calendar month, month, and year in which a diary week began, as well as a dummy for the day of week of payday, and a dummy for the second diary week, to control for survey fatigue. Z_i is a vector of socioeconomic and demographic controls, including household income, interest income, credit card ownership, age and occupation of the main earner, household size, number of income earners, marital status, geographic region of residence, and size of city of residence.

Each household member records spending for only fourteen days, making it necessary to pool observations in order to study expenditures over a whole month. Thus, at each distance from payday, we use diary weeks from many different households. Pooling household observations would lead to a biased estimate if characteristics are not orthogonal to distance from payday, but the FES survey takes place randomly during the year and across households. Furthermore, we verify the randomness of the survey empirically, in the next section.

Estimating the daily discount rate

Another goal of the paper is to estimate the percent decline in daily spending over the pay period, for the purposes of calibrating the standard and quasi-hyperbolic models.

¹⁵ Stephens (2004) identifies the payday effect using all weeks that overlap with payday. This confounds two possible effects – a payday effect and an end-of-pay period effect – which may reflect very different types of behavior.

With our data on seven-day aggregates of expenditure, observed at different distances from payday, we can estimate the implied profile of daily expenditure over the pay period, if we are willing to make an additional assumption. We make the simple assumption that daily expenditure declines linearly over the pay period. We do not expect this assumption to be strictly true, but it turns out to be a reasonable approximation given the weekly expenditure profile we observe, and we cannot reject the assumption statistically.

Assuming a linear decline of daily expenditure, the level of daily spending on day *t* of the pay period can be written:

$$e_t = E - \theta \cdot t + u_t \quad (2)$$

where *t* goes from zero until the day before the next payday, and θ describes the slope of the decline in daily spending. Given this expression for daily spending, and ignoring the error term for the moment, we can express the seven-day aggregate of expenditure beginning on day t, *a*_t, as:

$$a_{t} = \sum_{k=t}^{t+6} e_{k} = 7E - \theta t - \theta (t+1) - \theta (t+2) - \dots - \theta (t+6)$$
(3)

In the case of a 31-day pay period, this condition implies that the difference between any two overlapping, week-long aggregates of expenditure, which begin one day apart, can be written as:

$$a_{t+1} - a_t = -7 \cdot \theta \quad \text{for } 0 \le t \le 24 \tag{4}$$

$$a_{t+1} - a_t = 24 \cdot \theta \quad \text{for } 24 < t \le 30$$
 (5)

This says that dairy-week spending declines linearly over the first twenty-four days, with slope -7θ .¹⁶ We can thus recover θ , by estimating the average change in weekly expenditure over this portion of the pay period and then dividing by -7. Note that conditions (2) and (3) imply an asymmetric, "V-shape" for the weekly expenditure profile, which we can later compare to the profile actually observed in the data.¹⁷

To estimate the change in weekly expenditure over the pay period, we estimate a restricted, linear spline regression, where the first segment covers the first twenty-four days of the pay period, and the second covers the final six days. We impose the specific relationship between the slopes of these segments given by conditions (2) and (3). The resulting equation is:

$$c_{ii} = until_{2}4 + after_{2}4 + \pi \cdot (until_{2}4 \cdot \text{dist.} - (\frac{24}{7}) \cdot after_{2}4 \cdot \text{dist.} + \dots \quad (6)$$

Where the dependent variable is the *level* of weekly expenditure beginning on day *t*, *until_24* and *after_24* are separate intercepts for the two segments of the spline, and the key coefficient, π , is equal to -7θ by assumption. We estimate a similar regression for 30-day pay periods, where the low point of the spline occurs at distance 23.¹⁸ The regressions include all of the same controls as in our baseline specification.

¹⁶ Following the same line of reasoning, 30-day pay periods have a declining segment with slope $\pi = -7^*\theta$, up to distance 23, followed by a segment that increases with slope 23* θ over the rest of the pay period.

¹⁷ Figure 2 illustrates this relationship between daily and weekly spending with an example.

¹⁸ This portion of the analysis excludes 28-day and 29-day pay periods, as there are many fewer observations in these sub-samples.

Using our estimate of θ , and the median level of spending on the first day of the pay period, e_0 , we can then calculate the percentage change between each pair of adjacent days, and the average daily percent decline over the pay period.¹⁹ With an estimate for the percent decline in daily expenditure in hand, we are in a position to calibrate the standard and quasi-hyperbolic models and study daily time discounting.

Sample restrictions

The most significant sample restriction is the exclusion of people who are paid weekly. For people paid weekly, it is impossible to compare spending right after payday to spending before the next payday, because every observation includes either the last or the next payday.²⁰ We focus on the relationship between household consumption spending and distance from payday of the main earner. If there is more than one paycheck received by the household, on different paydays, this would tend to obscure the relationship of interest, so we drop households where there is any secondary earner whose paycheck is greater than 25% of total household wage earnings, *and* for whom the paycheck arrives 3 or more days away from the main earner's payday, or is not a monthly paycheck.²¹ We drop households missing information on key variables, households who

¹⁹ We solve for the median level of daily expenditure on payday using data on the median level of weekly spending, a_0 on payday: $a_0 = 7*E - \theta(1 + 2 + ... + 6) \implies E = (a_0 + 21*\theta)/7 = e_0$. By successively subtracting θ , we calculate the level of daily spending on all subsequent days of the pay period. We then express the changes in these levels in percentage terms, and calculate the average percentage decline in daily spending over the pay period. ²⁰ The sample includes roughly equal numbers of people paid monthly and people paid

 ²⁰ The sample includes roughly equal numbers of people paid monthly and people paid equally. About 90% of the sample are paid either weekly or monthly
 ²¹ While the precise values chosen for these cut-offs are somewhat arbitrary, the results

²¹ While the precise values chosen for these cut-offs are somewhat arbitrary, the results do not change appreciably with other cutoffs, e.g. eliminating households with secondary earners earning more than 33%, or more than 10%, of total household income.

have zero wage income, and households with a head who is retired or unemployed. We also drop outlier households with more than US \$5,000 of weekly consumption, or more than \$600 of weekly expenditures on highly non-durable goods.²² We use FES interviews conducted between 1988 and 2001. The FES extends back further than 1988, but issues of compatibility across years and the omission of key questions suggested the 1988 cutoff. Our final sample is composed of roughly 15,000 monthly-paid households. This translates into roughly 30,000 observations, because in most cases the sample includes two diary weeks for each household member.

Expenditures in the data are reported in pounds sterling. We adjust expenditures and pay amounts for inflation using the Retail Price Index for Britain, with 2000 as the base year.

3.3 Descriptive Statistics

Table 1 presents sample means of household characteristics, at different distances from payday. As expected given the survey design, these descriptive statistics appear to be orthogonal to distance, changing very little from one category to the next. Thus, although we include demographics in our regressions to check whether these variables affect the level of expenditure in a reasonable way, this is not strictly necessary for obtaining an unbiased estimate of the impact of distance.²³

²² About 250, or less than 1%, are excluded because of outlier values for total or highly non-durable consumption.

²³ We performed an additional check by successively adding demographics to the simplest regression with only the distance dummies and a constant term. Adding demographics did not lead to any appreciable change in the coefficient on distance, consistent with orthogonality.

Figure 3 presents frequency distributions for key variables: distance from payday, pay date, and diary start date. The first graph shows that distance from payday is evenly distributed, i.e. the timing of FES interviews and timing of paydays is roughly orthogonal. The second graph shows that pay dates are unevenly distributed; there is a strong concentration of pay dates on the last few days of the calendar month. The final graph in Figure 3 shows that diary start dates are fairly evenly distributed throughout the calendar month, as expected given the randomness of the FES interview during the year, with the exception of a dip in the number of observations around the 22^{nd 24}.

4. Results

4.1 Graphical Analysis

Figure 4 presents a plot of mean log expenditure versus distance from payday with 95% confidence bands.²⁵ Each point on the graph is calculated by averaging all week-long aggregates of expenditure that begin at that particular distance from payday. The vertical line indicates the weeks that are dropped from the sample to minimize measurement error.

Figure 4 shows that average consumption expenditures are markedly higher right after payday. Expenditure declines over the pay period, reaching a low around three weeks after payday, then starts to climb rapidly about a week before the next payday. The

²⁴ The dip is partly due to the concentration of paydays at the end of the month, and partly due to our exclusion of weeks starting 22 days from payday and beginning more than 4 days before the next payday.

²⁵ The log of zero observations on expenditure is undefined, so we assign these observations a value of zero in the graph. Dropping the zeros does not appreciably change the appearance of the graphs, which is not surprising given the small number of zero observations: we have only 482 observations with zero expenditure out of more than 30,000, or about 1.6% of the sample.

increase in weekly expenditure in the latter part of the pay period is presumably due to the fact that these weeks overlap with the next payday. The shape of the weekly expenditure profile also suggests that the "v-shape" implied by a linear decline in daily expenditure is not an unreasonable approximation.

4.2 Regression Analysis

Table 2 presents our baseline results, regressions of log weekly expenditure on distance from payday and controls.²⁶ For these regressions, we focus on the four distance categories discussed in Section 2. Regressions in Table 2, and all subsequent regressions, include our extensive array of demographic and time controls, but only the coefficients on distance categories are reported in the table. Coefficients from all demographic controls are presented in an appendix, available upon request. All regressions report robust standard errors, which are adjusted for clustering, i.e. possible correlation of the error term across diary weeks for the same household.²⁷

The first column of Table 2 shows that distance from payday has a significant impact on the category of all consumption spending. The dummy coefficient for 0 to 7 days after payday is significant at the 1% level, and says that consumption in diary weeks beginning in this interval is roughly 12% higher than for weeks starting 15 to 22 days after payday (the omitted distance category). This high level of spending extends well

²⁶ As mentioned in the previous footnote, zero expenditures make up a negligible fraction of the sample of diary week observations. As a check, we have estimated tobits for all regressions and find very similar results. We choose OLS as our preferred estimation method, mainly because it is easier to adjust standard errors for heteroscedasticity and clustering. For sub-categories of expenditures with more zero observations, however, we do report the tobit estimates.

²⁷ Except for the column 4 Tobit regressions, where doing such an adjustment is not trivial.

into the pay period: in weeks beginning 8 to 14 days after payday, spending is still higher than in the omitted category by about 5%. The 4 to 1 category captures the increase in spending observed at the very end of the pay period in Figure 4. The fact that the point estimate is even higher for the fourth category than for the 0 to 7 category suggests that the increase in weekly expenditure late in the pay period could reflect a combination of an end-of-pay-period effect, and the effect of the next payday. However, it is difficult to say exactly how much of the increase is due to each of these effects.

To verify that our results are not driven by our parameterization of the distance measure, we regress log expenditure on a set of dummy variables for each distance from payday. This non-parametric specification corroborates our baseline results: the individual distance dummies are highly significant and positive for the all consumption category, starting on payday and continuing until a distance of 13 days. These results are reported in full in Table B1 of Appendix B.

Table 3 presents results from the restricted spline regression. The first and second columns report π , the slope of the first segment of the spline, for 31- and 30-day pay periods. The absolute decline in daily expenditure, θ , is obtained by dividing these point estimates by -7, yielding a decline of -0.23 for 31-day pay periods and -0.10 for 30-day pay periods. For a household with the median level of expenditure on payday, this translates into a percentage daily decline between -0.6% to -0.2% (which gives an average monthly decline of 12%). Using F-tests that compare the fit of these restricted regressions to completely non-parametric regressions with separate dummy variables for each distance, we cannot reject the assumption of a linear daily decline. The estimates we

find using the assumption of linearity are quite similar to, although slightly smaller than, the 0.7% decline in food expenditures found by Shapiro (2004) using daily data.

The third column of Table 3 tests one possible explanation for why we find a smaller, and statistically insignificant π -coefficient for 30-day pay periods. In Figure 4, the data are reasonably close to a "v-shape," but in a graph for 30-day pay periods only (not shown), it is apparent that weekly expenditure begins to rise a bit earlier than predicted by a linear daily decline. Thus, the small point-estimate we find is likely to be attenuated, because it is based partly on a portion of the expenditure profile that is already increasing. The third column of Table 3 show that this is the case. Using an earlier threshold value of 21 leads to a steeper, and statistically significant, π -coefficient. One interpretation of the earlier rise in spending is that it reflects a systematic departure from our assumption of a linear decline, e.g. perhaps daily spending begins to rise just before the next payday. An alternative interpretation is that the early increase is due to the measurement error discussed above, in the timing of weeks during the pay period.

In summary, diary-week expenditure starts out significantly higher right after payday and then declines over most of the pay period, until it begins to rise rapidly again at the end. This pattern is robust to a non-parametric specification of the distance measure. The percentage daily decline implied by the data is similar to estimates from Shapiro(2004).

4.3 Robustness checks

Timing of non-discretionary payments

A possible explanation for declining expenditures between paydays is that the timing of non-discretionary payments, such as bills or pension contributions, happens to coincide with payday for most households. Returning to Table 2, however, the second column shows a significant decline in spending over the pay period, for spending that excludes bills and pension or savings contributions. This suggests that the payday effect is not driven by the timing of non-discretionary payments.

Intra-Household Strategic Motives

It is also possible that strategic interaction between household members could generate a downward sloping consumption path following payday. If household members compete for scare resources, then the optimal strategy could be for each household member to spend as much as possible, as fast as possible, right after payday in order to maximize their portion of the new inflow of resources. The third column of Table 2 tests this explanation by looking at the spending of single-person households (excluding bills). The decline in expenditures is particularly strong for this sub-sample, which goes against an explanation based on intra-household strategic interactions.²⁸

Stockpiling

Non-smooth consumption between paydays could also reflect household efforts to minimize transactions costs of shopping, by stockpiling durable goods. If the timing of this stockpiling happens to coincide with payday, it could explain a surge in expenditure

²⁸ Shapiro (2004) calls this the "intra-household scramble for resources." He finds that adding one additional member to the household does not affect the rate at which food stamp spending declines over the month.

right after payday. Given that the decline we observe extends well into the pay period, this explanation does not seem likely. However, the fourth column of Table 2 performs another test, showing the expenditure profile for the category of instant, or highly non-durable consumption goods.²⁹ The evident decline in spending on highly non-durable goods provides additional evidence that stockpiling is not a sufficient explanation for the payday effect we observe.

Calendar-Month Effects

Given that there is a concentration of paydays on the final days of the calendar month, we consider the possibility that some unobserved event, which regularly occurs at the end of the calendar-month, could also drive the increase in expenditure we observe. This concern is addressed by the inclusion of dummies for day-of-calendar-month in all regressions. However, in unreported regressions we take the additional precaution of estimating the expenditure profile over the pay period for the sub-sample of households who are paid in the interior of the calendar month. We find that the decline is just as strong and statistically significant for these households, mitigating concerns about confounding calendar-month effects.

In summary, we find little evidence to support several plausible explanations for the between-payday decline based on the standard model: monthly cycles in nondiscretionary payments, stockpiling, calendar-month effects, and strategic motives within the household.

²⁹ The category of instant consumption includes take-away food, alcohol and food consumed in bars and restaurants, cinema tickets, and admissions to discos.

5. Time Preference and Credit Constraints

One intuition for why households choose declining expenditures between paydays is that they are impatient, and face binding credit constraints.³⁰ This section evaluates this explanation with calibration exercises, assessing whether the standard model with exponential time preference, or a model with non-constant, hyperbolic time preference can better explain the magnitude of the decline we observe. We then test a prediction common to both models, that the decline should be present only for households who face binding credit constraints.

The Standard Model with Exponential Time Preference and Credit Constraints

It turns out that for a reasonable calibration of the standard model, time preference cannot explain a decline in expenditure over a short time horizon like a monthly pay period, even in the presence of binding credit constraints. Intuitively, this is because the rate of time preference, captured by the discount factor δ in the standard model, is constant, so that the δ needed to explain declining expenditure over a period of 30 days generates implausible predictions regarding longer-term impatience. We take this point from Rabin (2002), who illustrates the problem with a simple example: taking a *daily* rate of time preference of 1%, this implies a daily discount factor of 0.99 and an annual discount rate of 0.99365 = 0.03. This says that the consumer, who values consumption today 1% more than consumption tomorrow, must also value consumption today 97% more than consumption one year from now.

³⁰ Section A.1 of Appendix A provides a formal example, showing how, in the standard model, credit constraints lead to declining expenditures with a discontinuous jump after each payday.

To be more concrete, we perform a calibration exercise, calculating the exponential discount rate needed to rationalize our estimates of the decline in daily expenditure for the working population. The details of these calculations are shown in Section A.2 of Appendix A. We find that the daily discount rate needed to explain a decline in daily expenditure between -0.002 and -0.006 implies an annual discount factor in the range of 0.5 to 0.11. This says that the value a consumer places on consumption today must be between 65 and 89 percent greater than the value of consumption in one year's time. Such a degree of impatience over a year-long time horizon seems highly implausible, and illustrates the difficulty of explaining the decline in expenditures between month paydays with the constant rate of time preference in the standard model.³¹

A Model with Quasi-Hyperbolic Discounting and Credit Constraints

A model with hyperbolic discounting captures the intuition that consumers have a self-control problem arising from "present-biased preferences:" they plan to be frugal in the future, but cannot stick to their own plan, because their preferences put a special emphasis on present consumption. Non-constant discounting also makes it easier to explain a steep decline in spending over a short time period like a month, without implying implausible long-run impatience, because there is a separate discount rate for short- and long-run time horizons.

³¹ Alternatively, we could assume a more plausible discount rate of .8, which is still lower than standard empirical estimates, and calculate the implied elasticity of intertemporal substitution. We show in the appendix that this discount rate yields an implausibly large elasticity of substitution between 6 and 12 percent.

The quasi-hyperbolic model, a convenient approximation to a model with a hyperbolic discount function,³² adds one extra parameter, β , to the standard utility function:

$$U_t = E_t[u(c_t) + \beta \sum_{t=0}^T \delta^t u(c_{t+\tau})]$$
(7)

Where *t* indexes days, δ is equal to the discount factor in the standard model, and $\beta < 1$. In this case, the consumer has a short-term discount factor, $\beta \cdot \delta$, between today and tomorrow, that is relatively small compared to the discount factor of δ between any two future periods. This gives rise to the type of self-control problem described above; although the individual plans to be relatively patient in period *t*+1, discounting consumption in *t*+2 by only δ , once period *t* arrives he discounts *t*+2 by $\beta \cdot \delta$ and overspends from the perspective of his previous-period self.

We calibrate the quasi-hyperbolic model to see whether it can explain the decline we observe for the working population. In Section A.3 of Appendix A, we describe the optimal consumption path in the hyperbolic model, and explain in detail how we perform the calibration. What we find is that the hyperbolic model can explain an average rate of decline of -0.45, which is the midpoint of the range we observe. This is true assuming $\delta =$ *1*, and plausible values for the other free parameters in the model: $\beta = .93$, which is within the range of estimates found in laboratory experiments (Fredrick, Loewenstein and O'Donoghue, 2002), and an intertemporal elasticity of .45, which says that a 10 percent decrease in prices leads to a plausible, 4.5 percent increase in consumption.

³² The approximation was first proposed by Phelps and Pollak (1968) and has subsequently been used by Laibson (1994, 1997) and others.

Shapiro (2004) presents additional evidence on the role of time preference in household budgeting. He finds that consumers who appear impatient, in the sense that they would rather receive less than 50 dollars today than wait four weeks to receive a full 50 dollars, have a higher probability of having skipped a meal sometime during the previous month due to lack of money. Shapiro notes that reverse causality is potentially a problem, however, because a consumer who skipped a meal last month may still be desperate for money this month, not because of impatience, but because of the lingering ramifications of a negative shock to income last month.³³

Testing the role of credit constraints

We test whether binding credit constraints are a prerequisite for the observed decline in spending between paydays. Our first test uses asset interest-income as a proxy for liquidity or access to credit, which is the approach typically used in the consumption literature. Although this approach is standard, it is not entirely satisfactory; asset income is a rather indirect measure of the ability to smooth short-term consumption, and may also have an impact on behavior for reasons unrelated to its correlation with access to credit.³⁴ Our second test uses a more-direct and potentially superior measure of a

³³ Shapiro tackles this issue by controlling for the amount of food stamps remaining from the current month's allotment, and finds a smaller but still significant effect. But this does not entirely solve the problem. The reverse causality problem arises because food-stamp income may be insufficient for all of the household's food needs, making food consumption sensitive to negative shocks (food stamp income is presumably not subject to negative shocks). Thus it is necessary to control for current, total income rather than just food stamp income, in order to rule out lingering effects of a negative shock. ³⁴ We discuss one possible explanation below, in Section 7.

household's ability to smooth consumption: access to a credit card.³⁵ This measure comes from a question in the FES survey that asks, for each household member, whether that individual has access to a credit card.

Table 5 reports our results on credit constraints. The first two columns show no evidence of declining expenditures for households in the top quartile of asset income. However, the second two columns show evidence of a decline in expenditures for households with access to a credit card. The decline is significant, and only slightly more gradual than for non-cardholders. To rule out the unlikely possibility that most households with a credit card are fully "maxed-out" on their cards, and are thus effectively credit constrained, we estimate similar regressions (not shown) for households who report some positive spending with their credit card during the diary week. We find very similar results using these households.

In summary, we find mixed evidence supporting the prediction that credit constraints, in combination with non-constant time preference, drive the decline in expenditures. To the extent that high asset income is a proxy for credit constraints, our results match the prediction of the hyperbolic model. However, using access to a credit card, we find evidence of a strong decline. This suggests that something besides presentbiased preferences contributes to the decline in spending between paydays.

³⁵ Besides being farther-removed from the ability to smooth short-term consumption, high-asset income is probably a worse measure of the pure effects of borrowing constraints, due to selection: people in the top quartile of asset income are less likely to have self-control problems, or suffer from perceptual biases, which would attenuate the payday effect for reasons independent of credit constraints.

6. Method of Payment and Distance from Payday

The FES data distinguish between cash and credit card purchases, allowing us to investigate how households use these different methods of payment in relation to payday. For example, it might be that households only use their credit card as a last resort, to smooth consumption in pay periods when they have experienced a negative shock to income. In this case, we would expect to see that credit card expenditures increase over the pay period.

Table 6 looks at cash and credit card purchases separately, for households that have access to a credit card. We restrict the analysis to households who have used their credit card sometime during the diary week. The first two columns show that cash spending on all consumption goods exhibits a significant decline over the course of the pay period for these households. On the other hand, there is no statistically significant relationship between distance and expenditure. The second two columns show the same method-of-payment difference for spending excluding bills.³⁶ In unreported regressions, we find the same pattern for spending on instant consumption.

The method of payment difference we observe in monthly budgeting is suggestive of a system of mental accounting, in the sense of Thaler (1999). The idea of mental accounting is that individuals arbitrarily divide income into different mental categories, e.g. present income in the bank account, and future income accessible through credit cards, and have different rules for spending money from these "accounts." One

³⁶ Comparing cash and credit spending is complicated by the fact that that these methods of payment may be imperfect substitutes. If the timing of credit card purchases tends to be more non-discretionary, this could explain the lack of sensitivity to distance from payday. However, it is not obvious why the timing of credit card spending should be particularly non-discretionary, and furthermore, we find the same results when we exclude non-discretionary payments.

interpretation of the pattern we observe is that households are hyperbolic discounters, but have a rule against using their credit cards to make up for overspending at the beginning of the pay period with cash. Thus, although they are not credit constrained, they act as though they are, and end up with falling cash expenditures.³⁷ In the next section, we discuss another possible interpretation of the method of payment difference.

7. Discussion

One possible explanation for the decline in spending, which has not been discussed in the literature, is that households could have a custom of "celebrating" after each payday, with a binge in consumption. This practice could be related to culture, or class, which could explain the weaker decline observed for households with high assetincome. The celebration could also serve the function of a "self-reward," after another month of hard work. However, despite the intuitive appeal of this explanation, we find that the decline in spending extends well into the pay period, and thus seems too gradual to be consistent with a payday celebration or reward. This explanation also does not explain why there should be a celebration with cash spending but not with credit cards.

Another explanation corresponds to a common intuition regarding payday, which is that consumers have a tendency to feel that spending is "less costly" on payday, when there has been a salient increase in the bank balance. In this state, so the story goes,

³⁷ If there are two groups of cardholders, those who are maxed out on their credit cards, and thus have declining cash spending, and those who are not credit constrained and thus have essentially flat cash and credit spending, this could generate the type of pattern we observe. However, the results exclude households who make no use of their credit card, and furthermore, the decline in cash spending for cardholders is almost as steep as that of the sample as a whole, suggesting that it is not driven only by a small group of credit-constrained card holders.

consumers have trouble imagining that they will not always feel so "flush," and as a consequence they over-spend. Later, as the bank balance falls, consumers find that spending is increasingly painful.³⁸

This explanation could explain the extended decline we observe in the data, and has some basis in evidence from psychology. Experiments show that people have *reference-dependent preferences (RDP)*, i.e. they feel sensations of gains or losses when their income changes around a reference level. RDP are also characterized by *diminishing sensitivity (DMS)*: changes in income have a bigger impact on utility as distance from the reference level decreases. Supposing that consumers have a minimum, reference balance in mind during the pay period, e.g. a balance of zero, RDP could explain why consumers progress from a feeling of being wealthy to feeling of being poor over the course of a typical pay period.³⁹ In particular, DMS implies that spending (a negative change in the bank balance) has a smaller impact on RD utility when the paycheck first arrives, because the bank balance is far above the reference level, but becomes increasingly painful as the bank balance falls towards zero.⁴⁰

In Section A.4 of Appendix A, we outline a simple formal model that captures this intuition. We show that the consumer chooses declining expenditures, as long as the

³⁸ A similar intuition is present in the literature on budgeting behavior, e.g. Heath and Soll (1996) motivate their paper by arguing that self-imposed spending limits lead to artificial experiences of wealth and poverty, triggered by the amount of funds remaining in a given budget. Heath and Soll point to a passage from Sinclair Lewis' novel Babbit, which makes this point succinctly: "The effect of his scientific budget planning was that he felt at once triumphantly wealthy and perilously poor."

 ³⁹ RDP have also been used to explain savings and consumption behavior over the lifetime (Bowman, Minehart, and Rabin,1998) and portfolio choice (Thaler equity premium puzzle reference here).
 ⁴⁰This is similar in spirit to the "pain of paying" model of Prelec and Loewenstein (1998),

⁴⁰This is similar in spirit to the "pain of paying" model of Prelec and Loewenstein (1998), in which consumers experience an immediate hedonic cost when they spend income.

consumer makes the particular kind of mistake incorporated in the motivating intuition: the consumer expects the feeling that spending is painless on payday to last longer than it actually does.⁴¹

We also show that, although RDP can lead to declining expenditures between paydays, this can take the form of either over- or under-spending. To understand the intuition, consider a consumer who is fully forward looking, and calculates λ , the marginal utility of lifetime income in each period. If they also have RDP, this adds an additional, positive utility from having money in the bank, and thus makes them less willing to spend money at all points during the pay period. On the other hand, consider a consumer with RDP who does not calculate λ , in order to economize on cognitive costs.⁴² In this case, one possibility is that the consumer uses RD utility as a cognitively inexpensive, but flawed, technology for limiting spending. As a result, the consumer might overspends on payday, because the RD marginal valuation of income is close to zero, but cut back on spending later in the pay period, as the bank balance falls and spending becomes increasingly uncomfortable.⁴³

⁴¹ This type of mistake is well-supported by evidence from psychology, e.g. people who go shopping when they are hungry overspend, because they fail to predict that they will feel satiated in the near future (Gilbert, Gill, and Wilson, 1998). Loewenstein, O'Donoghue, and Rabin (2003) formalize this mistake in their model of *projection bias*.

⁴² The literature on bounded rationality emphasizes the (prohibitive) cognitive costs involved in calculating λ (Gabaix and Laibson, 2000), and surveys on budgeting behavior indicate that, in fact, many households do not have any sort of monthly budget or spending plan (Ameriks, Caplin, and Leahy, 2003a).

⁴³ Ameriks, Caplin, and Leahy (2003) have proposed a model of an "absent-minded" consumer, who avoids the cognitive costs of keeping track of spending by adopting a random purchasing strategy. The present argument is similar, in that a consumer seeks to avoid cognitive costs of budgeting, but differs in that the consumer has recourse to "budgeting by feel" rather than just random purchasing.

This model is admittedly speculative, but it has some intuitive appeal, and suggests an alternative interpretation of the evidence in this paper, and other papers on the payday effect. In this framework the gradual decline in spending is driven by a perceptual bias rather than impatience, and need not be limited to households facing binding credit constraints.⁴⁴ Given the assumption that the reference level does not increase with wealth, the model could explain why the decline becomes less pronounced for households with high asset-income: for these households, typical consumption expenditures may have a sufficiently small impact on the bank balance that the household never feels "poor" enough during the month to sharply curtail spending. The model also offers a different interpretation of the method-of-payment result: households may choose declining cash spending, because spending becomes increasingly painful as the bank balance falls, but choose flat credit card spending, because this type of spending does not have an impact on the current bank balance and always feels relatively painless.^{45,46}

It is possible that both impatience and perceptual biases could play a role in explaining the decline in spending between paydays, and assessing the relative

⁴⁴ It is noteworthy that the process of collecting the FES data effectively forces households to keep track of how much they spend, so that our results may understate the magnitude of the typical decline in the absence of such an intervention.

⁴⁵This interpretation was stated more eloquently by Russian playwright Anton Chekov: "When you live on cash you understand the limits of the world around which you navigate each day. Credit leads into a desert with invisible boundaries."

⁴⁶ An additional assumption is still needed, to explain why households do not choose increasing credit card spending over the pay period, as cash spending becomes more and more painful. Again, mental accounting is one possibility. Another possibility is that credit card spending is not a perfect substitute for cash spending, due to different transactions costs in different purchase situations. E.g. when one is running late to work, and standing at the head of a line of impatient customers, it is quicker and more pleasant to buy the morning coffee with cash than with a credit card. Imperfect substitutability between payment mediums, combined with the feeling that cash spending is painful but credit card spending is not, could generate the method of payment difference we observe.

importance of these factors seems a worthy subject for future research. For example, if overspending at the beginning of the pay period is driven by the salience of the increase in the bank balance, policies such as electronic direct deposit could help minimize the extent to which people "notice" the jump in the bank balance.

8. Conclusion

We find evidence of declining expenditures between paydays, for the typical household in the working population of the UK. This is hard to explain with the standard model, but is consistent with a quasi-hyperbolic model, which describes consumers who have trouble limiting their own spending because of "present-biased preferences." However, we find only mixed evidence supporting the prediction of the hyperbolic model, that binding credit constraints drive the decline. One possible explanation is suggested by the method-of-payment difference that we find: households could have a system of mental accounting, which limits the extent to which credit card spending is used to offset the fall in cash spending caused by impatience. We also discuss an alternative explanation, based on perceptual biases and cognitive costs of budgeting rather than impatience. More research is needed to disentangle the roles of impatience and perceptual biases, and to assess the magnitude of the welfare losses implied by sub-optimal budgeting.

References

- Akerlof, George A. (1991) "Procrastination and Obedience," *The American Economic Review*, 81(2), Papers and Proceedings of the Hundred and Third Annual Meeting of the American Economic Association, May, 1-19.
- Ameriks, John, Andrew Caplin and John Leahy (2003a) "Wealth Accumulation and the Tendency to Plan," *Quarterly Journal of Economics*, 118(3), 1007-1048.
- Ameriks, John, Andrew Caplin and John Leahy (2004) "The Absent-Minded Consumer," NBER working paper No. 10216.
- Attanasio, Orazio P. and James Banks (1998) "Trends in Households Saving: A Tale of Two Countries," *The Institute for Fiscal Studies (IFS)* Working Paper 98/15.
- Barenstein, Matias F. (2004) "Credit Cards and Consumption: An Urge to Splurge?" Unpublished, June.
- Benartzi, S. and R. H. Thaler (1995) "Myopic Loss Aversion and the Equity Premium Puzzle," *Quarterly Journal of Economics*, 110(1), 73-92.
- Bowman, David, Deborah Minehart, and Matthew Rabin (1998) "Loss Aversion in a Consumption-Savings Model," *Journal of Economic Behavior and Organization*, 38, 155-178.
- Browning, Martin and A. Lusardi (1996) "Household Saving: Micro Theories and Micro Facts," *Journal of Economic Literature*, 34(4), Dec., pp. 1791-1855.
- Burnham, Terry and Jay Phelan (2000) <u>Mean Genes: From Sex to Food, Taming our</u> <u>Primal Instincts</u>, New York, New York: Penguin.
- Camerer, Colin, Samuel Issacharoff, George Loewenstein, and Ted O'Donoghue (2003) "Regulation for Conservatives: Behavioral Economics and the Case for Asymmetric Paternalism," *University of Pennsylvania Law Review*, 151(3), January, 1211-1254.
- Carroll, Christopher D. (2001) "A Theory of the Consumption Function, With and Without Liquidity Constraints," *Journal of Economic Perspectives*, 15(3), Summer, pp. 23-45.
- Choi, James J., David Laibson, Brigitte C. Madrian, Andrew Metrick (2001) "Defined Contribution Pensions: Plan Rules, Participant Decisions, and the Path of Least Resistance," *NBER* Working Paper 8655.

- DellaVigna, Stephano and Ulrike Malmendier (2002) "Overestimating Self-Control: Evidence from the Health Club Industry," University of California, Berkeley, Manuscript.
- DellaVigna, Stephano and M. Daniele Paserman (2001) "Job-Search and Impatience," Harvard University, Manuscript.
- Fredrick, Shane, George Loewenstein and Ted O'Donoghue (2002) "Time Discounting and Time Preference: A Critical Review," *Journal of Economic Literature*, 40, 351-401.
- Gabaix, Xavier and David Laibson (2000) "Bounded Rationality and Directed Congnition," Working paper, MIT Dept. of Economics.
- Glink Ilyce (2001) <u>50 Simple Things You Can Do To Improve Your Personal Finances</u>, Three Rivers Press: New York.
- Gilbert, Daniel T., Michael J. Gill and Timothy D. Wilson (1998) "How Do We Know What We Will Like? The Informational Basis of Affective Forecasting," Working paper, Harvard Univ. Dept. of Psychology.
- Gross, David B. and Nicholas S. Souleles (2000) "Consumer Response to Changes in Credit Supply: Evidence from Credit Card Data," Financial Institutions Center Working Paper 00-04-B, The Wharton School, University of Pennsylvania (forthcoming QJE).
- Heath, Chip and Jack B. Soll (1996) "Mental Budgeting and Consumer Decisions," The Journal of Consumer Research, June, 23(1), 40-52.
- Hilton, George (1957) "The British Truck System in the Nineteenth Century," *The Journal of Political Economy*, 65(3), June, 237-256.
- Hsieh, Chang-Tai (2003) "Do Consumers React to Anticipated Income Changes? Evidence from the Alaska Permanent Fund," *The American Economic Review*, 93(1), March, pp.397-405.
- Jappelli, Tullio, Jorn-Steffen Pischke, and Nicholas Souleles (1998) "Testing for Liquidity Constraints in Euler Equations with Complementary Data Sources," *Review* of Economics and Statistics: 251-262.
- Interim Report (2003) Prime Minister' Strategy Unit on Alcohol Misuse, United Kingdom.
- Kahneman, Daniel and Amos Tversky (1979) "Prospect Theory: An Analysis of Decision Under Risk," *Econometrica*, 47, 263-291.

- Kahneman, Daniel and Amos Tversky (1992) "Advances in Prospect Theory: Cumulative Representations of Uncertainty, *Journal of Risk and Uncertainty*, Oct., 5, 297-324.
- Kahneman, Daniel and Amos Tversky, eds. (2000) <u>Choices, Values, and Frames</u>, New York: Russell Sage Foundation.
- Laibson, David (1996) "Hyperbolic Discount Functions, Undersaving, and Savings Policy," NBER Working Paper No. 5635.
- Laibson, David (1997) "Hyperbolic Discounting and Golden Eggs," *Quarterly Journal of Economics*, 112, 443-77.
- Laibson, David, Repetto, Andrea, and Jeremy Tobacman (1998) "Self Control and Saving for Retirement," *Brookings Papers on Economic Activity*.
- Laibson, David, Andrea Repetto, and Jeremy Tobacman (2000) "A Debt Puzzle," NBER Working Paper No. 7879.
- Loewenstein, G., Ted O'Donoghue and Matthew Rabin (2003) "Projection Bias in Predicting Future Utility," *Quarterly Journal of Economics*, 118, 1209-1248.
- Lusardi, Annamaria, Jonathan Skinner, and Steven Venti (2001) "Saving Puzzles and Saving Policies in the United States," *NBER* Working Paper 8237.
- O'Donoghue, T. and Matthew Rabin (1999a) "Doing it Now or Later," *American Economic Review*, March, 89(1), 103-24.
- O'Donoghue, T. and Matthew Rabin (2001) "Choice and Procrastination," *Quarterly Journal of Economics*, 116(1), 121-60.
- Ohls, James C., Thomas M. Fraker, Alberto P. Martini, and Michael Ponza (1992) "The effects of cash-out on food use by food stamp program participants in San Diego," Mathematica Policy Research, Inc.
- Osborn, Christabel (1898) "Economic Aspects of the Liquor Problem," *The Economic Journal*, 32(8), 572-577.
- Phelps, E. S. and R. A. Pollak (1968) "On Second-Best National Saving and Game Equilibrium Growth," *Review of Economic Studies*, 35, 185-99.
- Prelec, Drazen and George Loewenstein. (1998) "The Red and the Black: Mental Accounting of Savings and Debt," *Marketing Science*, 117(1), 4-28.
- Prelec, Drazen and Duncan Simester (2001) "Always Leave HomeWithout It: A Further Investigation of the Credit-Card Effect on Willingness to Pay," *Marketing Letters*, 12, 5-12.

- Rabin, Matthew (2000) "Risk Aversion and Expected-Utility Theory: A Calibration Theorem," *Econometrica*, 68(5), 1281-1292.
- Rabin, Matthew (2002) "A Perspective on Psychology and Economics," Forthcoming, *European Economic Review*.
- Read, Daniel, George Loewenstein and Matthew Rabin (1999) "Choice Bracketing," Journal of Risk and Uncertainty, 19:1-3, 171-197.
- Rizzo, John A. and Richard J. Zeckhauser (2003) "Reference Incomes, Loss Aversion, and Physician Behavior," *Review of Economics and Statistics*, 85(4), forthcoming.
- Shapiro, Jessie (2004) "Is There a Daily Discount Rate? Evidence from the Food Stamp Nutrition Cycle," Working paper, Harvard Univ. Dept. of Economics.
- Soman, Dilip (1997) "Contextual Effects of Payment Mechanism of Purchase Intention: Check or Charge," Working paper, Univ. of Colorado Dept. of Economics.
- Souleles, Nicholas S. (1999) "The Response of Household Consumption to Income Tax Refunds," *American Economic Review*, September: 947-958.
- Stephens, Melvin Jr. (2003) "3rd of Tha Month: Do Social Security Recipients Smooth Consumption Between Checks?" NBER Working Paper, #9135, forthcoming AER.
- Stephens, Melvin Jr. (2004) "Paycheck Receipt and the Timing of Consumption," NBER working paper Nr. 9356.
- Stephens, Melvin Jr. (2004) "The Consumption Response to Predictable Changes in Discretionary Income: Evidence from the Repayment of Vehicle Loans," Working paper, Carnegie Mellon Dept. of Economics.
- Thaler, Richard H (1999) "Mental Accounting Matters," Journal of Behavioral Decision Making, 12183-206.
- Tversky, Amos and Daniel Kahneman (1991a) "Loss Aversion in Riskless Choice: A Reference-Dependent Model," *Quarterly Journal of Economics*, 106, 1039-1061.
- Tversky, Amos and Daniel Kahneman (1992) "Advances in Prospect Theory: Cumulative Representations of Uncertainty," *Journal of Risk and Uncertainty*, 5, 297-323.
- Tversky, Amos and Daniel Kahneman (2000) *Choices, Values, and Frames*, Cambridge, MA: Cambridge University Press.
- Zeldes, Stephen P. (1989) "Consumption and Liquidity Constraints: An Empirical Investigation," *Journal of Political Economy*, 97(2), 305-345.

Figure 1: Data structure Overlapping Diary Weeks, or Seven-day Aggregates of Expenditure, by day of the pay period



Notes: Diary expenditures are aggregated over seven-day, or week-long periods in the data, which need not correspond to the calendar week. Instead, the data contain many overlapping "diary weeks," beginning at all different distances from payday. E.g. average "weekly" expenditure beginning on day 1 of the pay period would be calculated by averaging observations (1) and (3).

Figure 2: Linear Decline in Daily Consumption Expenditure and the Implied Dairy-Week Consumption Profile





Notes: The first panel in this figure simulates a particular linear decline in daily consumption expenditure. The second panel shows the implied, "V-shaped" profile of weekly expenditure, where each point in the figure represents a week-long aggregate, beginning at that distance from payday. Weekly expenditure begins to rise six days before the next payday, because these weeks overlap with the jump in spending on the next payday.



Figure 3: Descriptive Statistics

Notes: These panels show sample frequencies for key variables: distance from payday, pay date, and start date of a week-long expenditure diary.

Figure 4: Weekly Consumption Expenditure by Distance from Payday (With 95% confidence bands)



Notes: Each point in the figure represents an average of all diary weeks beginning at that distance. The vertical line indicates the exclusion of weeks beginning in the range of distances that are likely to be contaminated by measurement error, as discussed in Section 3.2 of the text.

_	Distance of diary start date from payday					
Mean Household						
	[-4 to -1 days	[0-7 days	[8-14 days	[15-22 days		
Characteristics	before]	after]	after]	after]		
A (1 1)	27.02	27.00	20.46	20.24		
Age (head)	37.83	37.90	38.46	38.34		
	(0.20)	(0.17)	(0.16)	(0.15)		
Income	1,727	1,710	1,694	1,694		
	(15.81)	(14.05)	(12.38)	(11.15)		
Credit Card	0.78	0.77	0.78	0.79		
	(0.01)	(0.00)	(0.00)	(0.00)		
Married	0.61	0.62	0.63	0.62		
	(0.01)	(0.01)	(0.01)	(0.01)		
No. of Earners	1.51	1.49	1.49	1.50		
	(0.01)	(0.01)	(0.01)	(0.01)		
Household Size	2.26	2.26	2.29	2.29		
	(0.01)	(0.01)	(0.01)	(0.01)		
Observations	5,465	7,359	8,384	9,219		

Table 1: Household Demographics by Distance from Payday

Notes: Standard errors in parentheses.

	All goods	No bills	No bills Single-person h.h.	Instant consumption		
	(1)	(2)	(3)	(4)		
0 to 7 days after	0.121	0.090	0.134	0.070		
	(0.013)***	(0.013)***	(0.031)***	(0.025)***		
8 to 14 days after	0.047	0.041	0.073	0.054		
-	(0.012)***	(0.011)***	(0.026)***	(0.023)**		
4 to 1 days before	0.149	0.108	0.133	0.095		
next payday	(0.014)***	(0.013)***	(0.032)***	(0.028)***		
Controls: All time, region, demographic controls, survey-fatigue dummy						
Observations	30424	30424	7021	30424		
Estimation method	OLS	OLS	OLS	Tobit		
R-squared	0.29	0.32	0.12	n.a.		

Table 2: Log Weekly Consumption as a Function of Distance fromPayday

Notes: Bills refer to payments with non-discretionary timing during the pay period, e.g. utility bill payments and pension contributions.. Model (3) is estimated for single-person households only. Instant consumption includes take-away food, alcohol and food consumed in bars and restaurants, cinema and disco tickets. Robust standard errors are in parentheses, corrected for clustering on household. * significant at 10%; ** significant at 5%; *** significant at 1%.

	All goods					
	31-day pay periods	30-day pay periods	30-day pay periods, (earlier threshold)			
	(1)	(2)	(3)			
Slope of first	-1.622	-0.722	-1.304			
segment.	$(0.434)^{***}$	(.624)	$(0.624)^{***}$			
e						
Implied absolute decline	$\theta = 0.233$	$\theta = 0.103$	$\theta = 0.186$			
Implied percent decline	-0.6%	-0.2%	-0.4%			
F-Test of restriction	<i>F</i> = 0.619	F = 0.526	F = 0.579			
Controls: All time, region, demographic controls, survey-fatigue dummy						
			•			
Observations	19218	10796	10796			
Estimation method	OLS	OLS	OLS			
R-squared	0.58	0.61	0.61			
*						

Table 3: Estimating the Decline in Daily Spending Level of Weekly Consumption as a Function of Distance from Payday Restricted Spline Regressions

Notes: These regressions impose a particular, asymmetric "V-shape" on the relationship between distance and weekly expenditure, implied by the assumption of a linear decline in daily spending. Section 3.2 explains how the absolute and percent declines in daily spending are inferred from the slope coefficient of the first segment. The first segment extends from day 1 to day 24 in model (1), 1 to 23 in model (2), 1 to 21 in model (3). The slope of the second segment of the spline is not reported, as it is restricted to be a particular multiple of the slope of the first segment. The model includes separate intercept terms for the two segments, which are also not shown. Robust standard errors are in parentheses, adjusted for clustering on household. * significant at 10%; ** significant at 5%; *** significant at 1%

	Top Quartile of Asset Income		Households With Credit Card			
	All goods	No bills	All goods	No bills		
	(1)	(2)	(1)	(2)		
0 to 7 days after	0.015	0.016	0.111	0.079		
	(0.032)	(0.031)	(0.015)***	(0.015)***		
8 to 14 days after	-0.008	-0.015	0.045	0.040		
	(0.029)	(0.026)	(0.013)***	(0.012)***		
4 to 1 days before	0.063	0.021	0.128	0.087		
next payday	(0.036)*	(0.033)	(0.016)***	(0.015)***		
Controls: All time, region, demographic controls, survey-fatigue dummy						
Observations	6068	6068	23706	23706		
Estimation	OLS	OLS	OLS	OLS		
method						
R-squared	0.24	0.26	0.26	0.29		

Table 4: Log Weekly Consumption on Distance from Payday,
by Alternative Measures of Access to Credit

Notes: This table reports separate regressions for households who are unlikely to be credit constrained: those with high asset income, and those with credit cards. Robust standard errors are in parentheses, adjusted for clustering on household. *significant at 10%; ** significant at 5%; *** significant at 1%

Table 5: Log Weekly Consumption on Distance from Payday,Cash Payment vs. Credit Payments of Cardholders.

	All goods]	No bills		
	Cash	Credit card	Cash	Credit card		
	(1)	(2)	(3)	(4)		
0 to 7 days after	0.107	-0.004	0.075	-0.015		
-	$(0.025)^{***}$	(0.048)	$(0.022)^{***}$	(0.048)		
8 to 14 days	0.044	0.047	0.038	0.047		
after	(0.023)**	(0.040)	$(0.02)^{*}$	(0.040)		
4 to 1 days before	0.115	-0.018	0.062	-0.023		
next payday	(0.027)***	(0.049)	(0.024)***	(0.049)		
Controls: All time, region, demographic controls, survey-fatigue dummy						
Observations	22140	22140	22140	22140		
Estimation method	Tobit	OLS	Tobit	OLS		
R-squared	n.a.	0.13	n.a.	0.13		

(conditional on non-zero credit card spending)

Notes: This table reports separate regressions for cash purchases and credit-card purchases, for households who have non-zero spending with a credit card during the diary week. Robust standard errors are in parentheses, adjusted for clustering on household. *significant at 10%; ** significant at 5%; *** significant at 1%

Appendix A

A.1 Time preference, credit constraints, and declining expenditures

In the standard model, a combination of impatience and credit constraints can generate a cycle of rising and falling expenditures with the peaks after each payday. Consider the benchmark model, in which a consumer maximizes a time-separable utility function with a constant discount factor $\delta < 1$ that reflects impatience. Assume that the consumer cannot borrow, and that there are two pay periods, with two periods each. The consumer thus maximizes utility subject to two budget constraints:

Max
$$U = \sum_{t=0}^{3} \delta^{t} u(c_{it})$$
 st. $Y_{0} \ge c_{0} + c_{1}$ and $Y_{3} + A \ge c_{3} + c_{4}$

In this problem, initial wealth is equal to income that arrives at the beginning of the first pay period, Y₀. Wealth in the third period is equal to the sum of savings from previous periods, *A*, and new income, *Y*₃, arriving in at the beginning of the second pay period (for simplicity, assume that the price of consumption equals 1 and that the interest rate is equal to zero). The resulting Euler equation $u'(c_t) = \delta u'(c_{t+1})$ implies $c_0 > c_1 > c_2 > c_3$, but the borrowing constraint causes the consumer to violate the Euler equation, i.e. *Y*₀ is not sufficient to allow $c_0 + c_1 > c_2 + c_3$. The consumer spends all she can on c_0 and c_1 because these yield more utility on the margin (due to impatience) than c_2 or c_3 , but then has income $Y_3 = Y_0$ left to spend on the remaining time periods (A = 0). Spending this income causes consumption to increase discontinuously at the beginning of the second pay period, back to the first-period level, $c_2 = c_0$ and then decline again to $c_3=c_1$.

A.2 Calibration of the standard, exponential model

Assume that the consumer has the same time-separable utility function as in the last sub-section, but maximizes utility over a thirty-day pay-period.⁴⁷ The consumer is credit constrained, and thus maximizes utility subject to Y_0 during the pay period:

Max
$$U = \sum_{t=0}^{T} \delta^{t} u(c_{it})$$
 st. $Y_0 \ge \sum_{t=0}^{T} \frac{p_t c_t}{R_t}$

Where period *t* is a day in a pay-period of length *T*, p_t is the price of consumption, and R_t is a gross interest rate. This maximization problem yields the standard Euler equation:

$$u'(c_{it}) = \frac{p_{t}}{p_{t+1}} \delta R u'(c_{it+1})$$
(8)

Assuming isoelastic utility, for which $u'(c_{ii}) = c_{ii}^{-\rho}$, one can take logs and express the Euler condition in terms of the percent daily change in consumption expenditure:

$$\ln(c_{it+1}) - \ln(c_{it}) = \frac{r - \gamma}{\rho} - \frac{\ln(p_{t+1}) - \ln(p_t)}{\rho}$$
(9)

Where $\gamma = -ln(\delta)$ is the daily discount rate, ρ is the inverse of the intertemporal elasticity of demand for consumption, and $r = ln(R_t)$ is the daily interest rate. At this point we make the assumption that prices are constant over the pay period, eliminating the second term in the expression.

We calculate the value of ρ implied by the percent decline in daily expenditure we report in Table 3, substituting conservative values for the other free parameters. We assume a value of 0.8 for the annual discount rate, which is well below typical estimates

⁴⁷ For purposes of comparison we follow the notation of Shapiro (2004) in the calibration exercises.

and implies a daily discount rate of 0.0006.⁴⁸ We also assume an annual interest rate of .05, which implies a daily interest rate of 0.0001. Our estimates of the daily decline range from -0.002 to -0.006, generating a value of ρ that implies an intertemporal elasticity of substitution between 4 and 12. These estimates suggest an implausibly large willingness to substitute consumption between days: using the midpoint of the lower range for the elasticity, a 10 percent decrease in the price on day *t*+*1* would lead to an increase in consumption of about 80 percent, relative to day *t*.

As an alternative exercise, we calculate the daily discount rate implied by the daily decline we observe, using a more reasonable value of $\rho = 1$, which corresponds to the case of log utility, and again assuming a daily interest rate of 0.0001. We find that percent decline in daily spending is compatible with a daily discount rate between - 0.0019 and -0.0061, but this generates an implausible annual discount factor between 0.5 and 0.11.

A.3 Calibration of the quasi-hyperbolic model

The quasi-hyperbolic model leads to the following generalized Euler equation, assuming isoelastic utility and assuming that the consumer is aware of their self-control problem (see Laibson, 1996 for a more complete derivation):

$$c_t^{-\rho} = [c'(W_{t+1})\beta\delta + (1 - c'(W_{t+1}))\delta]c_{t+1}^{-\rho}$$
(10)

Assuming an inability to borrow, the initial condition for calculating the optimal consumption path over a pay period T days long is consumption on the final day of the

⁴⁸ This is substantially lower than recent estimates that put the annual discount rate in the neighborhood of 0.9 or higher (Laibson, Repetto, and Tobacman, 2003; Gournichas and Parker, 2002).

pay period, $c_T = W_T$. Now suppose that, in a given period t+1, consumption is equal to $c_{t+1} = \alpha_{t+1}W_{t+1}$, and substitute into the generalized Euler equation:

$$c_{t}^{-\rho} = [\alpha_{t+1}\beta\delta + (1 - \alpha_{t+1})\delta](\alpha_{t+1}W_{t+1})^{-\rho} \quad (11)$$

Using the fact that $W_{t+1} = W_t - c_t$, and rearranging terms, we arrive at:

$$c_{t} = \frac{\alpha_{t+1}}{\alpha_{t+1} + (\delta(1 - (1 - \beta)\alpha_{t+1}))^{\frac{1}{\rho}}} W_{t} \quad (12)$$

Using this expression, and the initial condition $\alpha_T = I$, it is possible to solve recursively for the optimal consumption path.

We calibrate the hyperbolic model by assuming $\delta = I$ during the pay period, and then find values for β and ρ such that the consumption path yields a prediction in the observed range of -.006 to -.002 (the rate of decline is independent of the level of wealth W_t). We find that the model can predict the midpoint of the rage of estimates, -0.0045, for reasonable parameter values: $\beta = 0.93$ and an intertemporal elasticity of 0.45.

A.4 Model with Reference-Dependent Preferences and Projection Bias

A simple way to formalize the intuition discussed in Section 7 is to assume that the consumer maximizes a period *t* utility function:

$$u(c_t) + v(W_{t-1} - pc_t - r)$$
 (13)

Such that u() is a concave function of consumption, and v() is a function describing the hedonic value placed on distance of the current bank balance, W_t - pc_t, from a reference level r. Because v'() is assumed to be non-negative, spending p dollars causes a reduction in utility of -pv'(). The function v() also incorporates diminishing sensitivity, with the assumptions v''() < 0 if $W_t - pc_t > r$, and v''() > 0 if $W_t - pc_t < r$. For convenience, we

47

assume that the consumer never finds it optimal to go below *r* during the pay period.⁴⁹ The tendency for changes in the current bank balance to generate feelings of wealth or poverty is then a direct implication of diminishing sensitivity: v''(r) < 0 throughout the pay period, because the bank balance is never below *r*, and thus the hedonic cost of spending a dollar, -v'(r), increases as the bank balance falls.

Assume that the consumer suffers from projection bias, i.e. mistakenly expects the future valuation of income to be same as the current valuation, and faces a standard lifetime budget constraint without credit constraints, in which t indexes days and T is the length of the lifetime:

$$\sum_{t=0}^{T} pc_t = W_{t-1} + \sum_{t=0}^{T} Y_t . \quad (14)$$

In this case the consumer has the following spending rules for periods *t* and t+1, from the perspective of period *t*:

$$u'(c_{t}) - p \cdot v'(W_{t-1} - pc_{t} - 0) - p \cdot \lambda = 0 \quad (15)$$
$$u'(c_{t+1}) - p \cdot v'(W_{t-1} - pc_{t+1} - 0) - p \cdot \lambda = 0 \quad (16)$$

The consumer thus weighs the marginal utility of consumption against the standard marginal utility of income, λ , which is constant, and also against the reference-dependent marginal valuation of income v'(). Note that the consumer makes a mistake in period t (projection bias) by not taking into account the fact that current spending has an impact on reference-dependent utility in period t+1 and all other future periods, by reducing the

⁴⁹ Because we assume the individual does not go below r, *loss aversion*, the other key property of RDP, does not play a role. However, if we relax this assumption and include loss aversion the model makes the same qualitative predictions.

level of the bank balance. Once period t+1 arrives, the consumer finds that the true spending rule in t+1 is actually:

$$u'(c_{t+1}) - p \cdot v'(W_t - pc_{t+1} - 0) - p \cdot \lambda = 0 \quad (17)$$

Where $W_t = W_{t-1} - pc_t$. Given that v''(t) < 0, and $W_t < W_{t-1}$, it follows that the consumer chooses $c_t > c_{t+1}$, a declining consumption profile.⁵⁰

If the consumer makes a fully forward-looking calculation of λ , as we have assumed so far, the consumer chooses declining expenditures, but always *under-spends* because the marginal valuation of income is always greater than in the standard model: λ + v'(), with v'() > 0. An alternative assumption is that some consumers do not calculate λ , due to cognitive costs, in which theses consumers are likely to overspend on payday: the marginal valuation of income is simply v'(), which may be close to zero at the beginning of the pay period, implying that the consumer consumes to the point of satiation. As the pay period progresses, however, v'() increases and acts as a crude limit on spending.

⁵⁰ With partial projection bias, expenditure is still declining, as long as the bias is not too mild. More precisely, $c_t > c_{t+1}$ if $v'(Y_{t-1} - p c_t) < ((2\alpha - 1)/\alpha)v'(Y_t - pc_t + 1)$, where $0 < \alpha < 1$, and projection bias becomes complete as $\alpha \to 1$.

Appendix **B**

Table B1: Log Weekly Consumption vs. Distance from Payday, Non-ParametricDistance Specification

Expenditure			Instant	Instant
Diary Week	All goods	No bills	consumption	consumption
Starts:			consumption	consumption
	(1)	(2)	(3)	(4)
On payday	0.200	0.150	0.135	0.142
	(0.027)***	(0.025)***	(0.047)***	(0.051)***
1 Day After	0.130	0.082	0.024	0.026
	(0.028)***	(0.026)***	(0.047)	(0.050)
2 Days After	0.115	0.082	0.064	0.077
	(0.030)***	(0.027)***	(0.052)	(0.054)
3 Days After	0.126	0.107	0.096	0.106
	(0.030)***	(0.028)***	(0.052)*	(0.055)*
4 Days After	0.127	0.108	0.141	0.150
	(0.029)***	(0.027)***	(0.050)***	(0.053)***
5 Days After	0.150	0.118	0.026	0.029
	(0.029)***	(0.027)***	(0.051)	(0.053)
6 Days After	0.136	0.115	0.142	0.154
-	(0.030)***	(0.028)***	(0.053)***	(0.055)***
7 Days After	0.122	0.107	0.154	0.163
-	(0.027)***	(0.025)***	(0.048)***	(0.051)***
8 Days After	0.051	0.047	0.122	0.131
-	(0.028)*	(0.026)*	(0.049)**	(0.052)**
9 Days After	0.102	0.098	0.106	0.108
-	(0.031)***	(0.029)***	(0.054)*	(0.057)*
10 Days After	0.088	0.060	0.034	0.047
-	(0.029)***	(0.028)**	(0.051)	(0.055)
11 Days After	0.061	0.061	0.080	0.084
2	(0.028)**	(0.027)**	(0.048)*	(0.052)
12 Days After	0.096	0.066	0.047	0.052
2	(0.027)***	(0.025)***	(0.048)	(0.051)
13 Days After	0.058	0.071	0.142	0.152
2	(0.026)**	(0.024)***	(0.047)***	(0.050)***
14 Days After	0.033	0.046	0.086	0.089
2	(0.025)	(0.023)**	(0.043)**	(0.050)*
15 Days After	0.033	0.020	0.059	0.064
2	(0.026)	(0.024)	(0.046)	(0.051)
16 Days After	0.015	0.007	0.026	0.023
2	(0.028)	(0.026)	(0.051)	(0.053)
17 Days After	0.030	0.024	0.088	0.097
-	(0.028)	(0.026)	(0.050)*	(0.053)*
18 Days After	0.046	0.041	0.019	0.020

	(0.026)*	(0.025)*	(0.046)	(0.050)		
19 Days After	0.035	0.048	0.065	0.072		
-	(0.026)	(0.025)*	(0.045)	(0.049)		
20 Days After [†]	0.009	0.024	0.014	0.015		
	(0.026)	(0.024)	(0.047)	(0.050)		
-4 Days Before	0.167	0.127	0.078	0.079		
Payday	(0.027)***	(0.025)***	(0.047)*	(0.050)		
-3 Days Before	0.176	0.128	0.090	0.098		
	(0.025)***	(0.023)***	(0.043)**	(0.049)**		
-2 Days Before	0.170	0.148	0.167	0.183		
	(0.026)***	(0.025)***	(0.042)***	(0.051)***		
-1 Day Before	0.151	0.102	0.157	0.171		
	(0.029)***	(0.026)***	(0.051)***	(0.054)***		
Controls: All time, region, demographic controls, survey-fatigue dummy						
Observations	30424	30424	30424	30424		
Estimation	OLS	OLS	OLS	Tobit ¹		
method						
R-squared	0.29	0.32	0.17	n.a.		

Notes: The omitted distance category aggregates distances 21 and 22, due to low observations at these distances. Using 22 alone does not have an effect on the qualitative results in the table. Robust standard errors are in parentheses, adjusted for clustering on household. * significant at 10%; ** significant at 5%; *** significant at 1%.